Author's response to reviews

Title: Assessment of the role of transcript for GATA-4 as a marker of unfavorable outcome in human adrenocortical neoplasms

Authors:

Angela S Barbosa (angela.barbosa@hcnet.usp.br)
Luciano R Giacaglia (lucianogiacaglia@ig.com.br)
Regina M Martin (reginamm@usp.br)
Berenice B Mendonca (beremen@usp.br)
Chin J Lin (cjlin@bol.com.br)

Version: 8 Date: 28 June 2004

Author's response to reviews: see over
To Editorial Board  
BMC Endocrine disorders

Dear members of Editorial Board,

Please find attached the final version of manuscript by Barbosa et al entitled “Assessment of the role of transcript for GATA-4 as a marker of unfavorable outcome in human adrenocortical neoplasms” (MS# 1625451411217454). Minor modifications, as requested by Dr Peri and Editorial staff are listed below.

Sincerely,

Chin Jia Lin  
Laboratório de Hormônios e Genética Molecular LIM-42  
Hospital das Clínicas da FM-USP

As requested by Dr Peri, the following modifications has been introduced:

Inversion of references 6 and 7 has been corrected in both manuscript and Reference List.

The ‘.’ has replaced ‘,’ when numeric results was presented (pages 6, 9, 11).

Legends for Figures 6 and 7, inadvertently left in the previous version of manuscript, were removed.

Formatting changes requested by editorial staff:

Unnecessary capitalization from the title has been removed.

Email addresses for all authors have been included in the title page.

Affiliations have been written in full and provided in accordance to instructions given in your last email.

Title page has been created separating authors’ details from the rest of manuscript.

Reference list has been reformatted to conform to journal style.
All figures have been placed in portrait orientation. Unnecessary white space has been cropped out.